Author’s response to reviews

Title: Sample size requirements for case-control study designs

Authors:

Michael D deB Edwardes (michael.edwardes@clinepi.mcgill.ca)

Version: 5 Date: 2 Nov 2001

Just updated the abstract (should now be the same as in main text).
Updating first revision sent 2 days ago, along with replies to referees. Replies, appendix2, remain the same as 2 days ago. In main text, corrected minor typos and added a paragraph to the discussion, pointing to new developments.